Author's response to reviews

Title: Branchial cysts: an unusual cause of a mediastinal mass: a case report

Authors:

Vihar R Kotecha Dr (viharkotecha@hotmail.com)
Alex Muturi Dr (almuturi@gmail.com)
Josiah Ruturi Dr (ruturi@gmail.com)

Version: 2
Date: 26 March 2015

Author's response to reviews:

The title page: Full names of the authors have been included with indication of the corresponding author. Institutional address and emails of each author have been included.

Page 2: Abstract page 2
The abstract has been broken down into subsections namely introduction, case presentation and conclusion.

The consent section (page 4) has been updated to appear as per editors advise.

The references (page 5) have been re-written to meet the journal standard